9448007, 2023,

5, Do

/agupubs.onlinelibrary.wiley.com/doi/10.1029/2022GL101254 by Utrecht University Library, Wiley Online Library on [03/05/2023]. See the Terms.

and

use; OA art



Geophysical Research Letters[•]

RESEARCH LETTER

10.1029/2022GL101254

Key Points:

- Surface-wave velocity changes are directly linked to pore pressure variations through sensitivity kernels
- Pore pressure sensitivity kernels enable an inversion of surface-wave velocity changes for 4D pore pressure variations
- The shallow sensitivity to pore pressure changes in Groningen limits the method to the upper 200 m of the subsurface

Supporting Information:

Supporting Information may be found in the online version of this article.

Correspondence to:

E. Fokker, e.b.fokker@uu.nl

Citation:

Fokker, E., Ruigrok, E., Hawkins, R., & Trampert, J. (2023). 4D physics-based pore pressure monitoring using passive image interferometry. *Geophysical Research Letters*, 50, e2022GL101254. https://doi.org/10.1029/2022GL101254

Received 14 SEP 2022 Accepted 21 FEB 2023

© 2023. The Authors.

FOKKER ET AL.

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

4D Physics-Based Pore Pressure Monitoring Using Passive Image Interferometry

Eldert Fokker^{1,2} , Elmer Ruigrok^{1,2}, Rhys Hawkins^{1,3}, and Jeannot Trampert¹

¹Department of Earth Sciences, Utrecht University, Utrecht, The Netherlands, ²R&D Seismology and Acoustics, Royal Netherlands Meteorological Institute, De Bilt, The Netherlands, ³Now at School of Computing and Research School of Earth Sciences, Australian National University, Acton, ACT, Australia

Abstract This study introduces a technique for four-dimensional pore pressure monitoring using passive image interferometry. Surface-wave velocity changes as a function of frequency are directly linked to depth variations of pore pressure changes through sensitivity kernels. We demonstrate that these kernels can be used to invert time-lapse seismic velocity changes, retrieved with passive image interferometry, for hydrological pore pressure variations as a function of time, depth, and region. This new approach is applied in the Groningen region of the Netherlands. We show good recovery of pore pressure variations in the upper 200 m of the subsurface from passive seismic velocity observations. This depth range is primarily limited by the reliable frequency range of the seismic data.

Plain Language Summary In this study, we develop a method for pore pressure monitoring using seismic ambient noise. We use passive image interferometry to estimate surface-wave velocity changes as a function of frequency, and compute for surface-wave velocities the sensitivity to pore pressure changes as a function of depth. These so-called pore pressure sensitivity kernels are then used to invert surface-wave velocity changes for pore pressure variations as a function of depth. By comparing different regions of Groningen, The Netherlands, we build a four-dimensional pore pressure model for the shallowest 200 m of the subsurface. While the hydrological pore pressure variation can continue beyond 200 m depth, our method is limited by the shallow sensitivity and the frequency ranges for which seismic velocity measurements are possible.

1. Introduction

Traditionally, seismic imaging of the shallow subsurface is done with active sources. Seismic or acoustic sources from explosives or airguns excite downwards propagating waves, of which the reflections can be used to map geologic interfaces. Over the last decades, however, we have seen a shift toward passive imaging and monitoring. Seismic signals that were initially considered noise (e.g., microseisms) are now used to acquire subsurface data (e.g., Curtis et al., 2006).

Passive image interferometry (Sens-Schönfelder & Wegler, 2006) allows us to estimate seismic velocity changes using measurements of seismic ambient noise. This method consists of two steps. First, approximate Green's functions are estimated using cross-correlations of seismic noise measured at two receivers. This is referred to as seismic interferometry (Wapenaar, Draganov, et al., 2010). Second, velocity changes as a function of time are retrieved by comparing the coda of time-lapse cross-correlations to a reference. This step is referred to as coda wave interferometry (Lobkis & Weaver, 2003; Snieder, 2006). With passive image interferometry, a single lapse cross-correlation is generally constructed from noise measurements with a duration of a few hours to a few weeks, while the reference cross-correlation is often an average over one to a few years. The relative difference in arrival times dt/t then represents the relative velocity change dv/v = -dt/t with respect to the average reference velocity.

Seismic velocity variations have been empirically linked to many physical processes or observations, including temperature variations (e.g., Bièvre et al., 2018; Colombero et al., 2018; Richter et al., 2014), earthquake stress release (e.g., Brenguier et al., 2008; Sleeman & De Zeeuw-van Dalfsen, 2020; Wegler & Sens-Schönfelder, 2007), and hydrological stress fluctuations (e.g., Andajani et al., 2020; Clements & Denolle, 2018). For instance, Illien et al. (2022) used seismic velocity change and an empirical link with a hydrological model to find short-term permeability increases directly after earthquakes. Such empirical relationships can give very useful insights in the processes causing velocity changes, provided the empirical relationship reflect the physical processes involved. Therefore, we prefer a more physics-based approach.



10.1029/2022GL101254







Recently, Fokker et al. (2021) provided a physical model for pore pressure monitoring using surface-wave phase-velocity changes. Building on the theory of Tromp and Trampert (2018), they showed that pore pressure changes induce shear-wave velocity variations through changes in effective stress. Using surface-wave dispersion modeling (Hawkins, 2018), they showed that pore pressure changes explain the measured phase-velocity changes both in phase and amplitude.

In the current study, we demonstrate that measured velocity changes can be inverted for pore pressure variations as a function of time and space. We introduce pore pressure sensitivity kernels for surface-wave phase-velocity changes, and compute velocity variations by applying passive image interferometry to seismic ambient noise measurements in Groningen, The Netherlands. An inversion of these velocity changes results in models of pore pressure variation as a function of time, depth and region. Different regions of Groningen show a different temporal behavior that coincide with the jurisdictions of two independent water boards.

2. Groningen Setting, Data and Models

The Groningen region in the Netherlands has been studied extensively in the context of induced seismicity (e.g., Bourne et al., 2018; Hettema et al., 2017; Nepveu et al., 2016; Trampert et al., 2022) and subsidence (e.g., Van der Wal & Van Eijs, 2016; Van Thienen-Visser & Fokker, 2017; Van Thienen-Visser et al., 2015). The installation of a large dense network of borehole geophones (Dost et al., 2017) enabled intensive research activity. Seismic measurements on multiple depth levels were used to estimate shallow 1D velocity and attenuation profiles (Hofman et al., 2017; Ruigrok et al., 2022) and to estimate soil amplifications (Van Ginkel et al., 2019), while the large azimuthal coverage of the network was used to test different quality assessment parameters for passive image interferometry (Fokker & Ruigrok, 2019). The great amount of geological and geophysical models, provided by previous studies, and the presence of the large seismic network make Groningen an ideal region to test our approach of physics-based pore pressure monitoring.

The Groningen region can be divided into water board Noorderzijlvest in the northwest and water board Hunze en Aa's in the southeast. The borders between different water boards are shown in Figure 1 in light blue. Different water boards in the Netherlands can have different policies regarding groundwater management, and thus the pore pressure variations may be region dependent. In the southeastern region, at the location shown in Figure 1 as the blue dot, a deep borehole piezometer (Dinoloket, 2022) takes direct continuous measurements of the pore pressure at multiple depth levels up to 170 m. Shallow direct measurements of pore pressure variation can be found throughout the whole region (Grondwatertools, 2022).

Hydrologically, we can classify the shallow subsurface in the Groningen area roughly into three layers (Figure S1 in Supporting Information S1). In the first 25 m we find an unconfined aquifer. Pore pressure variations within this layer are a direct result of changes in the groundwater table. From 25 m to roughly 75–100 m, we find an aquitard, spanning the entire region with only sparse openings. Due to the low permeability of this clay layer, pore pressure diffusion cannot fully penetrate this layer and hence we do not expect large seasonal pore pressure variations. A confined aquifer can be found below the clay from 75–100 m to 200–300 m depth. The pore pressure in this layer is determined by the groundwater table at the recharge locations. Therefore, the spatial pore pressure variability is expected to be small within this layer.

From the seismic network in Groningen (Dost et al., 2017) we use data from the 4.5 Hz borehole geophones at 200 m depth at the locations shown in Figure 1 by the black triangles. We chose the deepest geophones from the borehole network, because they register the highest power of coherent noise from distant sources, compared to the power of incoherent noise from close sources. Each colored circle indicates a subregion that we investigate. For each subregion we gather shear-wave velocity and density models from Kruiver et al. (2017) and a compressional-wave velocity model from Romijn (2017). From these models we compute all elastic parameters needed in this study (Figure S2 in Supporting Information S1).

The models for compressional-wave velocity, shear-wave velocity and density (Figures S2a–S2c in Supporting Information S1) allow us to compute the bulk modulus, the shear modulus and the confining pressure (Figures S2d–S2f in Supporting Information S1). The pressure derivative of the shear modulus, needed for the sensitivity kernel, can be computed by a pointwise derivative of the shear modulus with respect to the confining pressure. At layer interfaces, however, the shear modulus can change abruptly due to a change in material from one layer to another. This will result in an unrealistic estimate for its pressure derivative. A smoothing operation with a robust weighing function and positivity constraint removes outliers that occur at such a layer intersection. Figure S2g in Supporting Information S1 shows our model for the pressure derivative of the shear modulus $d\mu/dp$ at the center of the corresponding region.

3. Passive Image Interferometry

To compute seismic velocity changes we apply passive image interferometry (Sens-Schönfelder & Wegler, 2006) to seismic ambient noise measured in Groningen, The Netherlands. This method consists of two processes. First, the Green's function between two seismic receivers is estimated using cross-correlations of ambient seismic noise. Second, time-lapse variations in arrival times are identified, corresponding to velocity variations.

To estimate the Green's function for one lapse period, we compute the cross-coherence of seismic noise, recorded by seismic receivers at locations x_A and x_B . The cross-coherence represents the spectrally normalized cross-correlation, and can be computed in the frequency domain (Wapenaar, Slob, et al., 2010):

$$\hat{H}(x_B, x_A, \omega) = \frac{\hat{u}(x_B, \omega)\hat{u}^*(x_A, \omega)}{|\hat{u}(x_B, \omega)||\hat{u}(x_A, \omega)|}.$$
(1)

where u is ground velocity. The frequency domain is indicated by a hat and the star denotes a complex conjugation. We stack cross-coherences calculated from 50% overlapping time windows of 20 min duration, where the first time window ranges from 0:00 to 0:20 UTC, the second from 0:10 to 0:30 UTC, etc., for a lapse period of 21 days. We repeat this procedure for lapse periods between 01 January 2017 and 01 January 2020. The cross-coherences are computed for vertical components. Figure S3 in Supporting Information S1 shows an example of cross-coherences in the time domain as a function of date, for receiver combination G014-G104 in the orange region (Figure 1) and frequency range [1.3 1.6] Hz.

We then determine velocity changes using the stretching method in the time domain (Lobkis & Weaver, 2003). Relative velocity changes $dv/v = \epsilon$ are found at the maximum correlation coefficient $CC(\epsilon)$ between lapse cross-coherence H_{lapse} , stretched in time with factor $(1 - \epsilon)$, and reference cross-coherence H_{ref} ,

$$CC(\epsilon) = \frac{\int_{t_1}^{t_2} H_{\text{lapse}}[t(1-\epsilon)] H_{\text{ref}}[t] dt}{\sqrt{\int_{t_1}^{t_2} (H_{\text{lapse}})^2 [t(1-\epsilon)] dt} \sqrt{\int_{t_1}^{t_2} (H_{\text{ref}})^2 [t] dt}}.$$
(2)

19448007, 2023, 5, Dov

1029/2022GI

.101254 by Utree

The reference cross-coherence is defined as the 3-year average from 01 January 2017 0:00 UTC to 01 January 2020 0:00 UTC, hence the retrieved velocity change is relative to the average within this period.

The coda of the cross-correlation is more likely to contain stable parts of the Green's function, because this only requires a stable background noise structure (Hadziioannou et al., 2009), while direct waves also require well-illuminated Fresnel zones (Wapenaar, Draganov, et al., 2010). For this reason, we omit all arrivals of direct waves, and choose our time windows (integration boundaries in Equation 2) for the cross-coherence as $\tau < |t| < 2\tau$, where $\tau = (x/v_{low} + 5)$ s. v_{low} is the fundamental-mode Rayleigh wave phase velocity in the model of Figures S2a–S2c in Supporting Information S1. An additional 5 s is added to exclude the direct Rayleigh waves with more certainty. This narrow window excludes most body waves in the coda and should mainly leave closely scattered surface waves.

We filter the cross-coherences with a bandpass filter before we estimate the velocity change for the chosen frequency range. To obtain velocity variations as a function of frequency range, we repeat this process for multiple frequency ranges. We compute an average velocity change for the regions indicated by the circles in Figure 1, using all receivers pairs within the indicated circles. This also allows us to compute the standard deviation of the sampling distribution of velocity change $\sigma_{dv/v} = \sigma/\sqrt{n}$, as an indication of the measurement uncertainty on the one hand, and the intrinsic variability over a region on the other hand.

We use the coda of the cross-coherence evaluated for the vertical components to estimate velocity changes. Likely, the velocity changes are caused by fundamental-mode Rayleigh waves, but contributions from higher modes, Love and body waves cannot a-priori be excluded. We repeat the approach of Fokker et al. (2021) to find what type of waves is the main contributor to the observed velocity change by making a forward calculation for the region containing the piezometer. Figure S4 in Supporting Information S1 shows velocity changes for five frequency ranges, retrieved using passive image interferometry (purple), and fundamental-mode phase-velocity changes for Rayleigh (red dashed) and Love (blue dashed) waves, modeled from the pore pressure variations measured by Dinoloket (2022). The velocity variations closely resemble fundamental-mode Rayleigh-wave velocity changes. Therefore, we treat the velocity changes measured on the vertical components as fundamental-mode Rayleigh-wave phase-velocity changes. We tried the same modeling with a Voigt average of Love and Rayleigh (Fokker et al., 2021), but this degraded the fit to the piezometer data.

4. Pore Pressure Sensitivity Kernels

To connect Rayleigh-wave phase-velocity change to pore pressure variation, we combine the physics-based relationship derived by Fokker et al. (2021) with shear-wave sensitivity kernels to construct pore pressure sensitivity kernels. Building on Tromp and Trampert (2018), Fokker et al. (2021) derived that a change in pore pressure u^0 via effective stress induces shear-wave velocity change

$$\frac{d\beta}{\beta} = -\frac{\mu'}{2\mu}u^0,\tag{3}$$

with shear-wave velocity β , shear modulus μ , and pressure derivative of the shear modulus $\mu' = d\mu/dp$. A positive change in pore pressure thus results in a negative change in shear-wave velocity.

Changes in the shear-wave velocity directly induce Rayleigh-wave phase-velocity changes

$$\frac{dv}{v}(\omega) = \int_0^\infty K_\beta(\omega, z) \frac{d\beta}{\beta}(z) dz, \qquad (4)$$

with Rayleigh-wave phase velocity v, and shear-wave sensitivity kernel K_{β} . We can now substitute Equations 3 in 4, resulting in

$$\frac{dv}{v}(\omega) = \int_0^\infty K_{u^0}(\omega, z) u^0(z) dz, \qquad (5)$$

where

$$K_{u^{0}}(\omega, z) = -\frac{\mu'(z)}{2\mu(z)} K_{\beta}(\omega, z)$$
(6)

governed by the applicable



Geophysical Research Letters



Figure 2. Visualization of Equation 6: (a) shear-wave sensitivity kernel $K_{\beta}(\omega, z)$ for Rayleigh-wave phase velocity, computed using the adjoint method (Hawkins, 2018) on models for compressional-wave velocity, shear-wave velocity and density (Figures S2a–S2c in Supporting Information S1; purple), (b) fraction $-\mu'(z)/2\mu(z)$ where μ is the shear modulus and μ' is the pressure derivative of the shear modulus (Figures S2e and S2g in Supporting Information S1; purple), and (c) pore pressure sensitivity kernel $K_{\mu^0}(\omega, z)$, which is a multiplication of figures (a) and (b). Note that the amplitude axes show logarithmic scales.

represents the pore pressure sensitivity kernel for Rayleigh-wave phase velocity.

Shear-wave sensitivity kernels for Rayleigh-wave phase velocity can be calculated using the adjoint method (Hawkins, 2018) together with one-dimensional models for compressional-wave velocity v_p , shear-wave velocity v_s , and density ρ . Figure 2a shows the shear-wave sensitivity kernel for the region centered at receiver G424 (purple region in Figure 1), constructed from the elastic model shown in Figures S2a–S2c in Supporting Information S1. The fraction $-\mu'/2\mu$ shown in Figure 2b is calculated using the shear modulus and its pressure derivative (Figures S2e and S2g in Supporting Information S1). In accordance with Equation 6, we multiply Figures 2a and 2b to obtain the pore pressure sensitivity kernel shown in Figure 2c.

5. Inversion for Pore Pressure Variation

To invert surface-wave velocity change for pore pressure variation as a function of depth and time, we need to discretize the linear relation described by Equation 5. We expand pore pressure change u^0 as

$$u^{0}(z,t_{k}) = \sum_{j} S_{j}(z)m_{j}(t_{k}),$$
(7)

where function $S_j(z)$ is chosen to be a cubic natural spline function, and $m_j(t_k)$ its coefficients at time t_k , which is the center of the 21 day lapse period (Section 3). We then rewrite Equation 5 as

$$\frac{dv}{v}(\omega_i, t_k) = \sum_j \int_0^\infty K_{u^0}(\omega_i, z) S_j(z) dz \, m_j(t_k). \tag{8}$$

For each lapse time t_k , this can be written as a linear forward problem,

$$\mathbf{d}(t_k) = \mathbf{G}\mathbf{m}(t_k),\tag{9}$$

where

$$d_i(t_k) = \frac{dv}{v}(\omega_i, t_k) \tag{10}$$

represents the data,

$$G_{ij} = \int_0^\infty K_{u^0}(\omega_i, z) S_j(z) dz$$
(11)

the forward operator, and $m_i(t_k)$ the model coefficients of the pore pressure change.



19448007, 2023, 5, Downloa

from https://aguput

loi/10.1029/2022GL101254 by Utrecht University Library, Wiley Online

_ibrary on [03/03/2023].

Model coefficients $m_i(t_k)$ can be retrieved using the explicit least-squares formulation (Tarantola, 2005),

$$\tilde{\mathbf{m}}(t_k) = \left(\mathbf{G}^T \mathbf{C}_{\mathbf{d}}^{-1}(t_k) \mathbf{G} + \mathbf{C}_{\mathbf{m}}^{-1}\right)^{-1} \mathbf{G}^T \mathbf{C}_{\mathbf{d}}^{-1}(t_k) \mathbf{d}(t_k),$$
(12)

with data covariance C_d and prior model covariance C_m . Based on the pressure head measurements in the southeastern region we expect a variance in pore pressure of 10⁶ Pa², hence we choose the model covariance as $C_m = 10^6 I$, where I represents the identity matrix. Since we are interested in the mean velocity change $dv/v(\omega_i, t_k)$ per region, we define the data covariance as the variance in the set of cross-coherences per region (see Figure 3a, error bars). We note that this variance can reflect the cross-coherence variability per region and/or direct observational uncertainty. We therefore use

$$\mathbf{C}_{\mathbf{d}}(t_k) = \operatorname{diag}\left(\boldsymbol{\sigma}_{\mathbf{dv/v}}(t_k)\right)^2. \tag{13}$$

The resolution $\mathbf{R}(t_k)$ of the inverted model representation $\tilde{\mathbf{m}}(t_k)$ can be obtained by substituting the data **d** in Equation 12 for the forward operator **G**,

$$\mathbf{R}(t_k) = \left(\mathbf{G}^T \mathbf{C}_{\mathbf{d}}^{-1}(t_k) \mathbf{G} + \mathbf{C}_{\mathbf{m}}^{-1}\right)^{-1} \mathbf{G}^T \mathbf{C}_{\mathbf{d}}^{-1}(t_k) \mathbf{G},$$
(14)

and the posterior model covariance can be found by

$$\mathbf{C}_{\tilde{\mathbf{m}}}(t_k) = \left(\mathbf{G}^T \mathbf{C}_{\mathbf{d}}^{-1}(t_k) \mathbf{G} + \mathbf{C}_{\mathbf{m}}^{-1}\right)^{-1}.$$
(15)

After inversion for model representation $m_j(t_k)$, we repeat the process for all lapse times t_k , and compute our final model for pore pressure variation using Equation 7.

Figure 3 shows the steps in the inversion scheme for the region centered at receiver G424 (purple region in Figure 1). Velocity changes retrieved using passive image interferometry form the data of this inversion (Figure 3a, error bars; two example frequency ranges). We use velocity variations of multiple frequency ranges with varying center frequency and frequency span (Figure 3b), and we define 10 spline functions S_j (Figure 3c). Following Equation 11, we construct forward operator G_{ij} (Figure 3d). Figure 3e shows pore pressure variations as retrieved using Equations 7 and 12, and Figure 3f shows the posterior model covariance as computed using Equation 15. The uncertainty of the retrieved model can then be computed using the square root of the diagonal of the posterior model covariance. Pore pressure changes smaller than this uncertainty are colored gray in Figure 3e. The resolution matrix is computed using Equation 14 (Figure 3g), indicating that we only have sufficient resolution to confidently infer the model coefficients corresponding to the first six splines. Therefore, pore pressure variations can only be retrieved at depths smaller than about 200 m. The resolution matrix shows that deeper pore pressure models have contributions from splines 2 and 6–10, and are thus smeared out over a large depth range. To show how well the pore pressure model explains the velocity variations, we use Equation 9, the forward operator **G**, and the inferred pore pressure model **m** to predict the data. Figure 3a (solid lines) shows the result.

We construct a four-dimensional pore pressure model by repeating the inversion procedure for all regions shown in Figure 1. We compute velocity changes (Figure 4a shows five example frequencies) and construct pore pressure sensitivity kernels based on the elastic parameters shown in Supporting Information S1 (Figure S2). The inversion leads to pore pressure models as a function of time, depth and region. Figure 4b shows in purple the inferred model in the region centered at receiver G424 for five depths, compared to the independent direct measurements of pore pressure variation in black (Figure 1, blue point; Dinoloket, 2022). The four-dimensional model of pore pressure variations is illustrated in Figure 4c, where for five depth levels and seven dates the pore pressure is shown in a colored map view. Detailed comparisons between pore pressure models and comparisons with shallow independent piezometric measurements are shown in Supporting Information S1 (Figures S5 and S6). The comparison of shallow pore pressure models in the northwest and the southeast shows significant spatial variations, while lateral variations of deeper pore pressure models could not be classified as significant. The shallow pore pressure models also compare well in phase and amplitude to the direct independent measurements of pore pressure change. The relative misfit between velocity change measured using passive image interferometry and predicted based on the inferred pore pressure model is shown in Supporting Information S1 (Figure S7), indicating that measured velocity variations between 0.7 and 1.8 Hz are well explained by our pore pressure model. In the lower frequency ranges, that is, larger depths, the model does not explain the data, in agreement with the information displayed on the posterior covariance and resolution matrix.

FOKKER ET AL.



Geophysical Research Letters



Figure 3. Inversion scheme for retrieving pore pressure variations: (a) seismic velocity changes as a function of time for two example frequency ranges, obtained using passive image interferometry (error bars), and predicted based on the inferred pore pressure model and the forward operator (solid lines), (b) all frequency ranges between 0.3 and 2 Hz for which velocity changes are computed, the frequencies in the pink band are excluded (see text), (c) 10 spline functions used to discretize pore pressure variations, (d) discretized pore pressure sensitivity kernel (i.e., forward operator G_{ij} in Equation 11, with spline functions as in (c), for the frequency ranges shown in (b)), (e) final model for pore pressure change as function of time and depth in accordance with Equations 7 and 12, (f) the posterior model covariance in accordance with Equation 15, and (g) resolution matrix in accordance with Equation 14.



Geophysical Research Letters



Figure 4. Four-dimensional variations in seismic velocity and pore pressure. The different colors indicate different regions, corresponding to the colors in Figure 1. (a) Seismic velocity change for five frequency ranges estimated using passive image interferometry (Sens-Schönfelder & Wegler, 2006) on the vertical components. (b) Inferred model for pore pressure variation in the region centered at receiver G424 for five depths. The black curves correspond to pore pressure measurements by the borehole piezometer indicated in Figure 1 as blue dot. (c) Map view of pore pressure models, as a function of time and depth. Each subplot corresponds to a certain time and depth, showing the pore pressure change as color for the seven different subregions presented in Figure 1.

6. Hydrologic Interpretation

The inferred pore pressure models reveal the characteristics of the hydrologic classification (Section 2, Figure S1 in Supporting Information S1).

Within the confined aquifer, pore pressure models compare well to the direct measurement in the southeast (Figure 4b) and models for the different regions are very similar to each other (Figures S5d–S5f in Supporting Information S1). The seasonal trends show lower pore pressures during summers and higher pore pressures during winters. The source for pore pressure change in this lower layer is due to locations where the clay layer is absent or very thin and pore pressure diffusion can reach this aquifer. Therefore, the pore pressure in this aquifer represents groundwater fluctuations at the recharge locations.

Within the aquitard, we observe small pore pressure variations that show neither a clear seasonal pattern, nor consistency over the different regions. Within this layer we expect much smaller pore pressure variations, because the hydraulic conductivity in the order of 1 mm per day is too low for pore pressure diffusion to reach the core of this layer. In the inversion process, pore pressure variations must therefore have leaked from depths corresponding to neighboring splines. The resolution in Figure 3g shows that this is possible.

Within the unconfined aquifer, pore pressure variations are a direct result of the changing groundwater table. Changes in the groundwater table are very site dependent, since their sources (i.e., precipitation, topography, groundwater extraction, and groundwater management) can vary from region to region. Interestingly, there is a significant (Figure S5 in Supporting Information S1) difference in amplitude between shallow pore pressure variations in the southeast (purple and blue areas) and the northwest (red and orange areas). Independent shallow piezometric measurements of the pore pressure (Grondwatertools, 2022) show for this aquifer an amplitude increase in seasonal variations from the southeast to the northwest. The amplitude differences between the regions coincide with the jurisdictions of two different water boards that may have different policies for groundwater management. The mismatch between shallow pore pressure models and the direct measurements shown in Figure 4b can potentially be explained by local topography or the presence of clay, since the direct measurements are taken at a point location, while the models represent an average over a lateral area of 250 km². The spatial variability shown by other pore pressure measurements from this region (purple area in Figure S6 in Supporting Information S1) supports this hypothesis. Other shallow pore pressure measurements (Grondwatertools, 2022) show closer agreement with the shallow models (Figure S6 in Supporting Information S1).

7. Discussion

In this study we obtained seismic velocity changes using the stretching method (Lobkis & Weaver, 2003). However, Zhan et al. (2013) showed that varying amplitudes in the noise can lead to spurious velocity changes. This is what we observe at frequency ranges containing the frequencies of 0.63 or 1.24 Hz, which are eigenfrequencies of nearby wind turbines (Van der Vleut, 2019). With varying wind direction, the swinging direction of the wind-turbine masts changes and therefore the directions, into which Rayleigh and Love waves are excited, will change. This causes substantial amplitude variations and hence spurious velocity changes. For this reason we excluded all frequency ranges containing these eigenfrequencies.

The advantage of the stretching method mostly lays in the ability to detect weak velocity changes using low signal-to-noise ratios. However, it makes use of the assumption of homogeneous velocity change. Using this method we can therefore only retrieve an average velocity change over a relatively large region. Alternatively, one could estimate velocity change using the moving window cross-spectral method (Clarke et al., 2011; James et al., 2017), dynamic time warping (Mikesell et al., 2015), or the wavelet method (Mao et al., 2020). These methods can be used for a higher-resolution spatial inversion of velocity change, taking into account the sensitivities of different wave types at different arrival times and frequencies (James et al., 2019; Mao et al., 2022; Margerin et al., 2016; Obermann et al., 2013).

By using the coda of the cross-correlations of vertical components close after the arrival of the fundamental-mode Rayleigh wave, we excluded most Love-wave energy. If the ratio of Love to Rayleigh energy were known in the Groningen area, one would be able to add velocity change measured on the horizontal components (i.e., RR, RT, TR, TT). The pore pressure sensitivity kernels for Rayleigh and Love would need to be averaged accordingly. A Voigt average between Rayleigh and Love as used by Fokker et al. (2021) would be too rough an approximation

for pore pressure inversion, since the ratio of Love to Rayleigh energy varies as a function of frequency (Juretzek & Hadziioannou, 2016).

Velocity changes are linked to pore pressure variations through pore pressure sensitivity kernels. To compute these kernels for Rayleigh-wave velocity change, we determined pressure derivatives of the shear modulus by a point-wise comparison between the shear modulus and the confining pressure. While this is a reliable method to determine the pressure derivative within a layer of one material, at interfaces this can lead to spurious values. A smoothing operation with a weighing function can remove such outliers at the cost of resolution. Alternatively, one could conduct a lab experiment to determine the pressure derivative of the shear modulus as a function of depth and hence maintain a better vertical resolution.

There are unexplained low-frequency data (Figure S7 in Supporting Information S1). For frequencies below 0.5 Hz we are pushing the 4.5 Hz geophones to their limits. With much instrumental noise at these frequencies, the retrieved velocity variations are of low quality. However, for the inversion part the quality of the low-frequency velocity variations does not really matter, since the resolution shows that the pore pressure models below 200 m cannot be interpreted anyway.

In this study we showed that the velocity variations between 0.7 and 1.8 Hz can be attributed to pore pressure changes. While in Groningen pore pressure change is the main source for velocity variation, other sources also need to be addressed. Locally, earthquakes can cause subsurface damage, resulting in a velocity drop (e.g., Brenguier et al., 2008; Wegler et al., 2009). However, this local effect has only been reported for much larger earthquakes than the ones observed in the Groningen area. Also temperature variations can induce seismic velocity changes (e.g., Colombero et al., 2018; Richter et al., 2014). Seasonal temperature variations by thermal diffusion through quartz, however, are naturally restricted to 0.1°C for depths below 20 m, and thermal energy storage systems only induce local temperature changes that cannot be resolved with our spatial resolution. Moisture variations within the vadose zone cause changes in density that can affect surface-wave velocities (e.g., Knight et al., 1998). In Groningen, however, the groundwater table can be found at approximately 1 m depth, which leaves a very small vadose zone and therefore a limited sensitivity to changes therein. For these reasons, we do not expect that other mechanisms should notably affect the seismic velocity, and therewith the pore pressure models at depths below 20 m.

Within the inversion procedure for depth variations of pore pressure, we used well-defined data and model covariances, enabling the use of the explicit Bayesian formulation. When data or model covariances are not available, it is still possible to carry out a damped least squares inversion. One can search for an optimum weight for the residual norm minimization and the solution norm minimization. Additionally, one could use the correlation coefficient $CC_{\max}(\omega, t)$ (Equation 2) as proxy for the quality of the retrieved velocity changes, since Fokker and Ruigrok (2019) showed that the standard deviation of retrieved velocity changes $\sigma(\omega_i, t_k)$ correlates strongly with $1 - CC_{\max}(\omega_i, t_k)$. Therefore, this can be used as an alternative to the data covariance presented in this study (Equation 13).

8. Conclusions

This study introduces a new technique for pore pressure monitoring using passive image interferometry. We derived that pore pressure sensitivity kernels can be used to link surface-wave velocity change as function of frequency directly to pore pressure change as function of depth. In Groningen, The Netherlands, most sensitivity to pore pressure changes lays in the very shallow subsurface (i.e., top 200 m), much shallower than the sensitivity to shear-wave velocity change. We showed that pore pressure sensitivity kernels can be used to invert surface-wave velocity changes for pore pressure variations as a function of depth, resulting in four-dimensional pore pressure models, agreeing with independent measurements of pore pressure variation and showing hydrological features.

Data Availability Statement

Seismic continuous data from the KNMI archive with Federation of Digital Seismograph Networks (FDSN) network identifier NL (KNMI, 1993, http://rdsa.knmi.nl/network/NL/) were used in the creation of this manuscript. Pressure head measurements are available through Dinoloket (2022, https://www.dinoloket.nl/en/subsurface-data)

Acknowledgments

DeepNL.2018.033.

We thank two anonymous reviewers

work is part of research programme

for their constructive comments. This

DeepNL, financed by the Dutch Research

Council (NWO) under project number

and Grondwatertools (2022, https://www.grondwatertools.nl/gwsinbeeld/). Models for shear-wave velocity, compressional-wave velocity and density were retrieved from Kruiver et al. (2017) and Romijn (2017). These models are available through https://osf.io/s3zxa/ (last accessed: 8 December 2022) and https://nam-onderzoeks-rapporten.data-app.nl/reports/download/groningen/en/3b4f8b0d-0277-40e0-8ff5-9a385c08327d (last accessed: 8 December 2022).

References

- Andajani, R. D., Tsuji, T., Snieder, R., & Ikeda, T. (2020). Spatial and temporal influence of rainfall on crustal pore pressure based on seismic velocity monitoring. *Earth Planets and Space*, 72(1), 1–17. https://doi.org/10.1186/s40623-020-01311-1
- Bièvre, G., Franz, M., Larose, E., Carrière, S., Jongmans, D., & Jaboyedoff, M. (2018). Influence of environmental parameters on the seismic velocity changes in a clayey mudflow (Pont-Bourquin Landslide, Switzerland). *Engineering Geology*, 245, 248–257. https://doi.org/10.1016/j.eng geo.2018.08.013
- Bourne, S. J., Oates, S. J., & Van Elk, J. (2018). The exponential rise of induced seismicity with increasing stress levels in the Groningen gas field and its implications for controlling seismic risk. *Geophysical Journal International*, 213(3), 1693–1700. https://doi.org/10.1093/gji/ggy084
- Brenguier, F., Campillo, M., Hadziioannou, C., Shapiro, N. M., Nadeau, R. M., & Larose, E. (2008). Postseismic relaxation along the San Andreas Fault at Parkfield from continuous seismological observations. *Science*, *321*(5895), 1478–1481. https://doi.org/10.1126/science.1160943
- Clarke, D., Zaccarelli, L., Shapiro, N., & Brenguier, F. (2011). Assessment of resolution and accuracy of the moving window cross spectral technique for monitoring crustal temporal variations using ambient seismic noise. *Geophysical Journal International*, 186(2), 867–882. https:// doi.org/10.1111/j.1365-246X.2011.05074.x
- Clements, T., & Denolle, M. A. (2018). Tracking groundwater levels using the ambient seismic field. *Geophysical Research Letters*, 45(13), 6459–6465. https://doi.org/10.1029/2018GL077706
- Colombero, C., Baillet, L., Comina, C., Jongmans, D., Larose, E., Valentin, J., & Vinciguerra, S. (2018). Integration of ambient seismic noise monitoring, displacement and meteorological measurements to infer the temperature-controlled long-term evolution of a complex prone-tofall cliff. *Geophysical Journal International*, 213(3), 1876–1897. https://doi.org/10.1093/gji/ggy090
- Curtis, A., Gerstoft, P., Sato, H., Snieder, R., & Wapenaar, K. (2006). Seismic interferometry—Turning noise into signal. *The Leading Edge*, 25(9), 1082–1092. https://doi.org/10.1190/1.2349814

Dinoloket (2022). Groundwater research; borehole identification B08C0952. Retrieved from https://www.dinoloket.nl/en/subsurface-data

Dost, B., Ruigrok, E., & Spetzler, J. (2017). Development of seismicity and probabilistic hazard assessment for the Groningen gas field. Netherlands Journal of Geosciences, 96(5), s235–s245. https://doi.org/10.1017/njg.2017.20

- Fokker, E., & Ruigrok, E. (2019). Quality parameters for passive image interferometry tested at the Groningen network. *Geophysical Journal International*, 218(2), 1367–1378. https://doi.org/10.1093/gji/ggz228
- Fokker, E., Ruigrok, E., Hawkins, R., & Trampert, J. (2021). Physics-based relationship for pore pressure and vertical stress monitoring using seismic velocity variations. *Remote Sensing*, 13(14), 2684. https://doi.org/10.3390/rs13142684
- Grondwatertools (2022). https://www.grondwatertools.nl/gwsinbeeld/
- Hadziioannou, C., Larose, E., Coutant, O., Roux, P., & Campillo, M. (2009). Stability of monitoring weak changes in multiply scattering media with ambient noise correlation: Laboratory experiments. *Journal of the Acoustical Society of America*, 125(6), 3688–3695. https://doi.org/ 10.1121/1.3125345
- Hawkins, R. (2018). A spectral element method for surface wave dispersion and adjoints. *Geophysical Journal International*, 215(1), 267–302. https://doi.org/10.1093/gji/ggy277
- Hettema, M. H., Jaarsma, B., Schroot, B. M., & Van Yperen, G. C. (2017). An empirical relationship for the seismic activity rate of the Groningen gas field. Netherlands Journal of Geosciences, 96(5), s149–s161. https://doi.org/10.1017/njg.2017.18
- Hofman, L., Ruigrok, E., Dost, B., & Paulssen, H. (2017). A shallow seismic velocity model for the Groningen area in The Netherlands. Journal of Geophysical Research: Solid Earth, 122(10), 8035–8050. https://doi.org/10.1002/2017JB014419
- Illien, L., Sens-Schönfelder, C., Andermann, C., Marc, O., Cook, K. L., Adhikari, L. B., & Hovius, N. (2022). Seismic velocity recovery in the subsurface: Transient damage and groundwater drainage following the 2015 Gorkha earthquake, Nepal. Journal of Geophysical Research: Solid Earth, 127(2), e2021JB023402. https://doi.org/10.1029/2021JB023402
- James, S., Knox, H., Abbott, R., Panning, M., & Screaton, E. (2019). Insights into permafrost and seasonal active-layer dynamics from ambient seismic noise monitoring. *Journal of Geophysical Research: Earth Surface*, 124(7), 1798–1816. https://doi.org/10.1029/2019JF005051
- James, S., Knox, H., Abbott, R., & Screaton, E. (2017). Improved moving window cross-spectral analysis for resolving large temporal seismic velocity changes in permafrost. *Geophysical Research Letters*, 44(9), 4018–4026. https://doi.org/10.1002/2016GL072468
- Juretzek, C., & Hadziioannou, C. (2016). Where do ocean microseisms come from? A study of Love-to-Rayleigh wave ratios. Journal of Geophysical Research: Solid Earth, 121(9), 6741–6756. https://doi.org/10.1002/2016JB013017
- Knight, R., Dvorkin, J., & Nur, A. (1998). Acoustic signatures of partial saturation. *Geophysics*, 63(1), 132–138. https://doi.org/10.1190/1.1444305
 KNMI. (1993). Netherlands seismic and acoustic network. Royal Netherlands Meteorological Institute (KNMI), Other/Seismic Network. https://doi.org/10.21944/e970fd34-23b9-3411-b366-e4f72877d2c5
- Kruiver, P. P., Van Dedem, E., Romijn, R., de Lange, G., Korff, M., Stafleu, J., et al. (2017). An integrated shear-wave velocity model for the Groningen gas field, The Netherlands. *Bulletin of Earthquake Engineering*, 15(9), 3555–3580. https://doi.org/10.1007/s10518-017-0105-y
- Lobkis, O. I., & Weaver, R. L. (2003). Coda-wave interferometry in finite solids: Recovery of P-to-S conversion rates in an elastodynamic billiard. *Physical Review Letters*, 90(25), 254302. https://doi.org/10.1103/PhysRevLett.90.254302
- Mao, S., Lecointre, A., Van der Hilst, R. D., & Campillo, M. (2022). Space-time monitoring of groundwater fluctuations with passive seismic interferometry. *Nature Communications*, 13(1), 1–9. https://doi.org/10.1038/s41467-022-32194-3
- Mao, S., Mordret, A., Campillo, M., Fang, H., & Van der Hilst, R. D. (2020). On the measurement of seismic traveltime changes in the timefrequency domain with wavelet cross-spectrum analysis. *Geophysical Journal International*, 221(1), 550–568. https://doi.org/10.1093/gji/ ggz495
- Margerin, L., Planès, T., Mayor, J., & Calvet, M. (2016). Sensitivity kernels for coda-wave interferometry and scattering tomography: Theory and numerical evaluation in two-dimensional anisotropically scattering media. *Geophysical Journal International*, 204(1), 650–666. https:// doi.org/10.1093/gji/ggv470
- Mikesell, T. D., Malcolm, A. E., Yang, D., & Haney, M. M. (2015). A comparison of methods to estimate seismic phase delays: Numerical examples for coda wave interferometry. *Geophysical Journal International*, 202(1), 347–360. https://doi.org/10.1093/gji/ggv138

- Nepveu, M., Van Thienen-Visser, K., & Sijacic, D. (2016). Statistics of seismic events at the Groningen field. *Bulletin of Earthquake Engineering*, 14(12), 3343–3362. https://doi.org/10.1007/s10518-016-0007-4
- Obermann, A., Planès, T., Larose, E., Sens-Schönfelder, C., & Campillo, M. (2013). Depth sensitivity of seismic coda waves to velocity perturbations in an elastic heterogeneous medium. *Geophysical Journal International*, 194(1), 372–382. https://doi.org/10.1093/gji/ggt043
- Richter, T., Sens-Schönfelder, C., Kind, R., & Asch, G. (2014). Comprehensive observation and modeling of earthquake and temperature-related seismic velocity changes in northern Chile with passive image interferometry. *Journal of Geophysical Research: Solid Earth*, 119(6), 4747–4765. https://doi.org/10.1002/2013JB010695
- Romijn, R. (2017). Groningen velocity model 2017 (Technical Report). Nederlandse Aardolie Maatschappij NV. Retrieved from https://namfeitenencijfers.data-app.nl/download/rapportdialog/9a5751d9-2ff5-4b6a-9c25-e37e76976bc1
- Ruigrok, E., Rodriguez-Marek, A., Edwards, B., Kruiver, P., Dost, B., & Bommer, J. (2022). Derivation of a near-surface damping model for the Groningen gas field. *Geophysical Journal International*, 230(2), 776–795. https://doi.org/10.1093/gji/ggac069
- Sens-Schönfelder, C., & Wegler, U. (2006). Passive image interferometry and seasonal variations of seismic velocities at Merapi Volcano, Indonesia. *Geophysical Research Letters*, 33(21), L21302. https://doi.org/10.1029/2006GL027797
- Sleeman, R., & De Zeeuw-van Dalfsen, E. (2020). Cross-correlation analysis of long-term ambient seismic-noise recordings in the Caribbean Netherlands to monitor the volcanoes on Saba and St. Eustatius. Bulletin of the Seismological Society of America, 110(5), 2541– 2558. https://doi.org/10.1785/0120200011
- Snieder, R. (2006). The theory of coda wave interferometry. Pure and Applied Geophysics, 163(2), 455–473. https://doi.org/10.1007/s00024-005-0026-6
- Tarantola, A. (2005). Inverse problem theory and methods for model parameter estimation. Society for Industrial Mathematics. https://doi. org/10.1137/1.9780898717921
- Trampert, J., Benzi, R., & Toschi, F. (2022). Implications of the statistics of seismicity recorded within the Groningen gas field. Netherlands Journal of Geosciences, 101, e9. https://doi.org/10.1017/njg.2022.8
- Tromp, J., & Trampert, J. (2018). Effects of induced stress on seismic forward modelling and inversion. *Geophysical Journal International*, 213(2), 851–867. https://doi.org/10.1093/gji/ggy020
- Van der Vleut, B. C. M. (2019). Characterizing the seismic signal of individual wind turbines in the Borgsweer area (Master's thesis). Utrecht University. Retrieved from https://studenttheses.uu.nl/handle/20.500.12932/33586
- Van der Wal, O., & Van Eijs, R. (2016). Subsidence inversion on Groningen using levelling data only. NAM report EP201612206045 (Technical Report). Nederlandse Aardolie Maatschappij NV. Retrieved from https://nam-feitenencijfers.data-app.nl/download/rapport/30346ae1-8443-452f-96a6-314ee5c91d41?open=true
- Van Ginkel, J., Ruigrok, E., & Herber, R. (2019). Assessing soil amplifications in Groningen, The Netherlands. First Break, 37(10), 33–38. https://doi.org/10.3997/1365-2397.2019026
- Van Thienen-Visser, K., & Fokker, P. A. (2017). The future of subsidence modelling: Compaction and subsidence due to gas depletion of the Groningen gas field in The Netherlands. *Netherlands Journal of Geosciences*, 96(5), s105–s116. https://doi.org/10.1017/njg.2017.10
- Van Thienen-Visser, K., Pruiksma, J., & Breunese, J. (2015). Compaction and subsidence of the Groningen gas field in The Netherlands. Proceedings of the International Association of Hydrological Sciences, 372, 367–373. https://doi.org/10.5194/piahs-372-367-2015
- Wapenaar, K., Draganov, D., Snieder, R., Campman, X., & Verdel, A. (2010). Tutorial on seismic interferometry: Part 1—Basic principles and applications. *Geophysics*, 75(5), 75A195–75A209, https://doi.org/10.1190/1.3457445
- Wapenaar, K., Slob, E., Snieder, R., & Curtis, A. (2010). Tutorial on seismic interferometry: Part 2—Underlying theory and new advances. *Geophysics*, 75(5), 75A211–75A227. https://doi.org/10.1190/1.3463440
- Wegler, U., Nakahara, H., Sens-Schönfelder, C., Korn, M., & Shiomi, K. (2009). Sudden drop of seismic velocity after the 2004 Mw 6.6 mid-Niigata earthquake, Japan, observed with passive image interferometry. *Journal of Geophysical Research*, 114(B6), B06305. https://doi. org/10.1029/2008JB005869
- Wegler, U., & Sens-Schönfelder, C. (2007). Fault zone monitoring with passive image interferometry. *Geophysical Journal International*, 168(3), 1029–1033. https://doi.org/10.1111/j.1365-246X.2006.03284.x
- Zhan, Z., Tsai, V. C., & Clayton, R. W. (2013). Spurious velocity changes caused by temporal variations in ambient noise frequency content. Geophysical Journal International, 194(3), 1574–1581. https://doi.org/10.1093/gji/ggt170